Author's response to reviews

Title: Archetype Relational Mapping - A practical openEHR Persistence Solution

Authors:

li wang (daowangli@icloud.com)
Lingtong Min (ericmin1987@zju.edu.cn)
Rui Wang (13834579858@163.com)
Xudong Lu (lyxd@zju.edu.cn)
Huilong Duan (duanhl@zju.edu.cn)

Version: 8 Date: 7 October 2015

Author's response to reviews: see over
Dear Editors:

Thanks for your review comments. The comments are very important for us to improve our paper, the revised parts according to the comments as follows:

1. Update email address of Rui Wang to 13834579858@163.com

We are very grateful for your warm work earnestly. In all, we found the comments are quite helpful. They point the deficiencies about our manuscript, also the aspects that we have not done enough, and helps us for further improvement.

Thank you and best regards.

Yours sincerely,

Li Wang

Corresponding author:

Name: Xudong Lu

E-mail: lvxd@zju.edu.cn